\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{https://ijrps.com}{\color{blue}\underline{\smash{https://ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\begin{document}


\def\RunningAuthor{Allehabi Muhaisn A et al.}
\firstPage{2351}
\articleType{Original Article}
\receivedDate{03.04.2019}
\acceptedDate{15.07.2019}
\revisedDate{10.07.2019}
\journalVolume{10}
\journalIssue{3}
\journalDoi{ijrps.v10i3.1478}
\copyrightYear{2019}
\def\authorCount{5}
\def\affCount{5}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{Exploring the status of treating chronic pain among elderly patients hospitalized in internal medicine wards in saudi arabia}
\author{Allehabi~Muhaisn~A\textsuperscript{*}\textsuperscript{1},  
            Alshammari~Mohammad~A\textsuperscript{2},  
            Alenzy~Muhya~T\textsuperscript{3},  
            ALshehri~Mohammed~A\textsuperscript{4}, 
            Almusri~Ali~D\textsuperscript{5}~\\[5pt]{\textsuperscript{1}Taif Unversity, Taif, Saudi Arabia}~\\{\textsuperscript{2}Prince Mohammed Medical City, Riyadh, Saudi Arabia}~\\{\textsuperscript{3}Department of Pharmacy Practice, Faculty of Pharmacy\unskip, Princess Nourah bint Abdul Rahman University, Riyadh, Saudi Arabia}~\\{\textsuperscript{4}Khamis Mushayt General Hospital, Khamis Mushayt, Saudi Arabia}~\\{\textsuperscript{5}King Khaled Hospital,  Najran, Saudi Arabia}}

\begin{abstract}
The elderly population experience many health issues that range from emotional, psychological to physiological conditions. Previous studies reveal that chronic pain is among the major health issues affecting old people. The processes of managing chronic pain among the elderly have proven to be challenging practices since individuals exhibit different health complications in advanced age. Saudi Arabia is among the countries that have made progressive improvements to the approaches employed to address chronic pain. Thus, exploring the status of elderly patients experiencing chronic pain in three major hospitals in Saudi Arabia is an educational strategy to understand the situation both in the middle east and the rest of the world. The notable chronic ailments associated with the elderly in Saudi Arabia include diabetes, back pain, and depression. Several studies have shown that the systems for treating older adults that suffer from chronic pain in Saudi Arabia are effective. The study used the 20-item pain anxiety symptom scale (PASC) to asses chronic pain for 53 patients draw from King Fahad Hospital in Jeddah, Asir Central Hospital, and King Khaled hospital in Najran. The study revealed that chronic pain causes fear. Inability to think properly. Concertation problem and lack of sleep, among other issues. Apparently. The study borrows from other studies done outside Saudi Arabia to explore the issue problem in a broader spectrum. The understanding of chronic pain status is an inevitable exercise towards establishing effective pain management programs or systems. 
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Chronic Pain,\newline Treating,\newline Internal Medicine,\newline Elderly Patients,\newline Pain Anxiety Symptom Scal
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Allehabi~Muhaisn~A~\\ Phone:\ 0535994518~\\  Email:\ tmalenzy@pnu.edu.sa
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{https://ijrps.com}{https://ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
Chronic pain also acknowledged as persistent pain, which lasts within an individual for more than 3 months \unskip~\citep{582558:13491382}. The study aimed at exploring the status of treating chronic pain amid the elderly patients, who are hospitalised in Saudi Arabia's internal medicine department. Evidence suggests that in this regard, the elderly patients hospitalised in internal medicine in Saudi Arabia mainly suffer pain from various diseases, including depression and diabetes, among others. Based on the reported 12-month occurrence rates among the hospitalised elderly patients in Saudi Arabia, it has been apparent that 7\% to 28\% and 4\% to 17\% of them face chronic pain, which is mainly due to depression and anxiety respectively \unskip~\citep{582558:13491382}. In Saudi Arabia, it is projected that the numbers of elderly people would increase in the future, thereby raising the scope of more hospitalisation in the internal medicine department \unskip~\citep{582558:13491383}. The elderly patients belonging to Saudi Arabia are mainly hospitalised in the internal medicine department with health disorder with respect to low back pain (LBP). This particular health issue is likely to affect 25.1\% and 35.1\% of men and women correspondingly in Saudi Arabia \unskip~\citep{582558:13491385}. 

Based on the observation made by \unskip~\citep{582558:13491384}, the status of treating chronic pain amid the elderly patients hospitalised in Saudi Arabia's internal medicine department is effective. This is owing to the reason of improved healthcare system, increased use of effective medicines, and quality training provided to the nurses regarding treatment as well as care management. Apart from the medicines, usage of herbal treatment cannot also be ignored while analysing the status of treating chronic pain among the elderly patients, who are admitted in the kingdom's internal medicine department \unskip~\citep{582558:13491381}. 

\textbf{Ethics Approval }

\textbf{\space }For this study, all procedures were in accordance with the ethical standards of the Scientific Research Ethics Committee for Saudi Universities and with the 1964 Helsinki declaration and its lateramendment.
    
\section{Materials and Methods}
\textbf{Research Methodology }

To explore the status of treating chronic pain among elderly patients hospitalized in internal medicine wards in Saudi Arabia, the researcher will use the following tools Figure~\ref{f-00f2e615573b}.


\bgroup
\fixFloatSize{images/5dc0b10e-c3fb-4419-a679-1491fb3fbb4c-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/5dc0b10e-c3fb-4419-a679-1491fb3fbb4c-upicture1.png}{\includegraphics{images/5dc0b10e-c3fb-4419-a679-1491fb3fbb4c-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Research Design}}
\label{f-00f2e615573b}
\end{figure}
\egroup
\textbf{Questionnaire}

 A questionnaire will be developed to retrieve some information from the elderlies staying at medical wards. The informationFigure~\ref{f-00f2e615573b} will be related to: 


  
  \begin{enumerate}
  \item \relax Age (65-74 or 75-84).
  \item \relax Education.
  \item \relax Marital Status (Single, Married, Widowed).
  \item \relax Living arrangement (Alone, with a spouse, with sons, with spouse and sons, other).
  \item \relax BMI (body mass index) at admission.
  \item \relax SL (severity index) at admission.
  \item \relax CT (comorbidity type) at admission.
  \item \relax BI at admission.
  \item \relax Psychological status at admission.
  \item \relax Duration of hospital stay.
  \item \relax Status of the patient at discharge (Discharge, discharge critical condition, transferred, dead).
  \item \relax Type of drugs provided.
  \item \relax The total number of drugs.
  \end{enumerate}
  
\bgroup
\fixFloatSize{images/5c1501ac-f2ff-4992-829e-d2f7f0c6a805-ucapture.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/5c1501ac-f2ff-4992-829e-d2f7f0c6a805-ucapture.png}{\includegraphics{images/5c1501ac-f2ff-4992-829e-d2f7f0c6a805-ucapture.png}}{}
\makeatother 
\caption{\boldmath {Q1 I~can't think straight when in pain}}
\label{f-8d66b1254e10}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/7e4b1448-80ad-46f9-b7d9-c0d73a59ac2f-ucapture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/7e4b1448-80ad-46f9-b7d9-c0d73a59ac2f-ucapture3.png}{\includegraphics{images/7e4b1448-80ad-46f9-b7d9-c0d73a59ac2f-ucapture3.png}}{}
\makeatother 
\caption{\boldmath {Q2 During painful episodes it is difficult for me to think of anything besides the pain?}}
\label{f-709f6476c85d}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/da6d69cf-7406-4aca-850d-057e7cab2e7d-ucapture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/da6d69cf-7406-4aca-850d-057e7cab2e7d-ucapture4.png}{\includegraphics{images/da6d69cf-7406-4aca-850d-057e7cab2e7d-ucapture4.png}}{}
\makeatother 
\caption{\boldmath {Q3 When I~hurt, I~think about pain constantly}}
\label{f-eb9b53a10e94}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/60ba265d-084d-4f5e-8620-b2e6d610a72d-ucapture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/60ba265d-084d-4f5e-8620-b2e6d610a72d-ucapture5.png}{\includegraphics{images/60ba265d-084d-4f5e-8620-b2e6d610a72d-ucapture5.png}}{}
\makeatother 
\caption{\boldmath {Q4 I~find it hard to concentrate when I~hurt}}
\label{f-15f9209c0693}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/48d363ab-87ca-444f-8dc3-aa21b2854a69-ucapture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/48d363ab-87ca-444f-8dc3-aa21b2854a69-ucapture6.png}{\includegraphics{images/48d363ab-87ca-444f-8dc3-aa21b2854a69-ucapture6.png}}{}
\makeatother 
\caption{\boldmath {Q5 I~worry when I~am in pain}}
\label{f-90b135027fad}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/9aa58d72-6402-48c1-a50e-0ab6a1ac18d9-ucapture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/9aa58d72-6402-48c1-a50e-0ab6a1ac18d9-ucapture7.png}{\includegraphics{images/9aa58d72-6402-48c1-a50e-0ab6a1ac18d9-ucapture7.png}}{}
\makeatother 
\caption{\boldmath {Q6 I~go immediately to bed when I~feel severe pain}}
\label{f-c79278136b94}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/f58c39c7-afbb-4bea-aaa4-4ef8bd927ce3-ucapture8.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/f58c39c7-afbb-4bea-aaa4-4ef8bd927ce3-ucapture8.png}{\includegraphics{images/f58c39c7-afbb-4bea-aaa4-4ef8bd927ce3-ucapture8.png}}{}
\makeatother 
\caption{\boldmath {Q7 I~will stop any activity as soon as I~sense pain coming on}}
\label{f-2dcfe5b7ce9f}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/461162eb-8dd4-4226-8f87-71bbbf6d81f1-ucapture9.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/461162eb-8dd4-4226-8f87-71bbbf6d81f1-ucapture9.png}{\includegraphics{images/461162eb-8dd4-4226-8f87-71bbbf6d81f1-ucapture9.png}}{}
\makeatother 
\caption{\boldmath {Q8 as soon as Pain comes on, I~take medication to reduce It}}
\label{f-060d35c5beb0}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/9a1631dc-382c-4a65-9898-42ecc6e5a8ab-ucapture10.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/9a1631dc-382c-4a65-9898-42ecc6e5a8ab-ucapture10.png}{\includegraphics{images/9a1631dc-382c-4a65-9898-42ecc6e5a8ab-ucapture10.png}}{}
\makeatother 
\caption{\boldmath {Q9 I~avoid important activities when I~hurt}}
\label{f-98e1c8767acb}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/91fb7a44-cbc8-4952-bf04-53272b039642-ucapture11.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/91fb7a44-cbc8-4952-bf04-53272b039642-ucapture11.png}{\includegraphics{images/91fb7a44-cbc8-4952-bf04-53272b039642-ucapture11.png}}{}
\makeatother 
\caption{\boldmath {Q10 I~try to avoid activities that cause pain}}
\label{f-98b8932ea974}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/91bbb0fe-db91-458a-ba73-3a682e7fe7e8-ucapture12.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/91bbb0fe-db91-458a-ba73-3a682e7fe7e8-ucapture12.png}{\includegraphics{images/91bbb0fe-db91-458a-ba73-3a682e7fe7e8-ucapture12.png}}{}
\makeatother 
\caption{\boldmath {Q11 I~think that if my pain gets too severe it will never decrease}}
\label{f-b9d11dae0d4e}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/9c9c605a-9fea-4ee9-91ec-6157eb3e1318-ucapture13.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/9c9c605a-9fea-4ee9-91ec-6157eb3e1318-ucapture13.png}{\includegraphics{images/9c9c605a-9fea-4ee9-91ec-6157eb3e1318-ucapture13.png}}{}
\makeatother 
\caption{\boldmath {Q12 When I~feel pain, I~am afraid that something terrible will happen}}
\label{f-b6ce7dd3733e}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/70b60b05-dded-4613-a461-88ef19d56dcf-ucapture14.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/70b60b05-dded-4613-a461-88ef19d56dcf-ucapture14.png}{\includegraphics{images/70b60b05-dded-4613-a461-88ef19d56dcf-ucapture14.png}}{}
\makeatother 
\caption{\boldmath {Q13 When I~feel pain, I~think I~might be seriously ill}}
\label{f-0ad4bf2bebd6}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/c67e804d-a2a0-4c60-8e4f-3835e3529e41-ucapture15.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/c67e804d-a2a0-4c60-8e4f-3835e3529e41-ucapture15.png}{\includegraphics{images/c67e804d-a2a0-4c60-8e4f-3835e3529e41-ucapture15.png}}{}
\makeatother 
\caption{\boldmath {Q14 Pain sensations are terrifying}}
\label{f-768e35ce19f6}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/00eec7dd-61aa-49c9-8403-1f87910367f0-ucapture16.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/00eec7dd-61aa-49c9-8403-1f87910367f0-ucapture16.png}{\includegraphics{images/00eec7dd-61aa-49c9-8403-1f87910367f0-ucapture16.png}}{}
\makeatother 
\caption{\boldmath {Q15 When pain comes on strong, I~think that I~might become paralyzed or more disabled}}
\label{f-6676691511e1}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/152cdd01-3451-48cc-bb92-901d3d97c9b5-ucapture17.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/152cdd01-3451-48cc-bb92-901d3d97c9b5-ucapture17.png}{\includegraphics{images/152cdd01-3451-48cc-bb92-901d3d97c9b5-ucapture17.png}}{}
\makeatother 
\caption{\boldmath {Q16 I~begin trembling when engaged in activity that increases pain}}
\label{f-35da03c3ddec}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/dd0cae98-3783-4b3e-8eb5-996bb56433b7-ucapture18.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/dd0cae98-3783-4b3e-8eb5-996bb56433b7-ucapture18.png}{\includegraphics{images/dd0cae98-3783-4b3e-8eb5-996bb56433b7-ucapture18.png}}{}
\makeatother 
\caption{\boldmath {Q17 Pain seems to cause my heart to pound or race}}
\label{f-3cc45f2e4cd2}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/c52ca567-a82a-4cc0-bf86-2b7492b9b8f1-ucapture19.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/c52ca567-a82a-4cc0-bf86-2b7492b9b8f1-ucapture19.png}{\includegraphics{images/c52ca567-a82a-4cc0-bf86-2b7492b9b8f1-ucapture19.png}}{}
\makeatother 
\caption{\boldmath {Q18 When I~sense pain, I~feel dizzy or faint}}
\label{f-72a173e6753a}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/a537d271-d077-4e51-9aac-34e5d25d4eb9-ucapture20.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/a537d271-d077-4e51-9aac-34e5d25d4eb9-ucapture20.png}{\includegraphics{images/a537d271-d077-4e51-9aac-34e5d25d4eb9-ucapture20.png}}{}
\makeatother 
\caption{\boldmath {Q19 Pain makes me nauseous}}
\label{f-31023b0820c0}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/66c3920d-cb01-46fd-b2e5-03c8dc04a041-ucapture21.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/66c3920d-cb01-46fd-b2e5-03c8dc04a041-ucapture21.png}{\includegraphics{images/66c3920d-cb01-46fd-b2e5-03c8dc04a041-ucapture21.png}}{}
\makeatother 
\caption{\boldmath {Q20 I~find it difficult to calm my body down after periods of pain}}
\label{f-42d573a8cd90}
\end{figure}
\egroup
\textbf{Interviews}

semi-structured interviews will be conducted with both patients (elderlies) and medical services providers (nurse staff and physicians). The information will be related to


  
  \begin{enumerate}
  \item \relax The common type of pain amongst elderlies.
  \item \relax The level of pain amongst elderlies.
  \item \relax The frequency of visiting the hospital.
  \item \relax The level of medical services provided.
  \end{enumerate}
  \textbf{Research Sample}

Three big different hospitals in Saudi Arabia (King Fahad Hospital in Jeddah, Asir central Hospital and King Khaled hospital in Najran).
    
\section{Results and Discussion}
The status of chronic pain treatment among 53 hospitalized elderly Saudi patients was assessed using the 20-item pain anxiety symptom scale (PASC).  When asked whether the pain affects their straight-thinking, 41.51 \% responded sometimes, 30.19 \% said usually, and 13.21\% always responded; with 9.43\% responding and 5.66\% never (Figure~\ref{f-8d66b1254e10}). When further asked whether they think about nothing else other than pain during painful episodes, a majority still sometimes responded (43.40 \%), usually (28.30 \%) and always 15.09 \%; with the minority rarely responding (9.43 \%) or never (3.77 \%) (Figure~\ref{f-709f6476c85d}). The responses were similar to the above items when alternatively, asked whether when hurt, they constantly thought about pain, with majority sometimes responding (41.51 \%), usually (33.96 \%) and always 9.43 \%; with the minority rarely responding (9.43 \%) or never (5.66 \%) (Figure~\ref{f-eb9b53a10e94}). However, when further asked whether they find it hard to concentrate when hurt, the majority reported usually (39.62 \%), always or sometimes (24.53 \%), with a minority responding never or rarely (5.66 \%) (Figure~\ref{f-15f9209c0693}). After asking patients about pain-precipitated anxiety, a majority responded that they get worried when in pain, with most patients usually responding (32.08 \%), always (26.42 \%) or sometimes (22.64 \%) and the least rarely responding (13.21 \%) or never (5.66 \%) (Figure~\ref{f-90b135027fad}). This pattern of pain-precipitated worry appeared to influence their decision regarding going to bed immediately when there is severe pain. In this case, the majority sometimes responded (28.30 \%), usually or always (26.42 \%) with the minority rarely responding (13.21 \%) or never (5.66 \%) (Figure~\ref{f-c79278136b94}). The responses regarding going to bed when in severe pain appeared inconsistent with those regarding the decision to stop their activities upon sensing an impending pain. In that case, a substantial majority responded sometimes (41.51 \%), rarely (16.98 \%) or usually (11.32 \%) with substantial minority responding always/never (15.09 \%) or (16.98 \%) (Figure~\ref{f-2dcfe5b7ce9f}). By contrast, the responses regarding going to bed when in severe pain were, however, consistent with their responses on taking pain-relieving medication as soon as pain ensues; with a substantial majority sometimes responding (32.08 \%), always (15.09 \%) or usually (16.98 \%) with substantial minority reporting rarely (20.75 \%) or never (15.09 \%) (Figure~\ref{f-060d35c5beb0}). Responses on the decision to stop their activities upon sensing an impending pain were consistent with responses regarding avoiding important activities when hurt, with majority sometimes responding (39.62 \%), always (16.98 \%) or usually (15.09 \%); and the minority rarely responding (13.21 \%) or never (15.09 \%) (Figure~\ref{f-98e1c8767acb}). These are further consistent with responses regarding avoidance of activities that cause pain, with the majority usually responding (35.85 \%), always (22.64 \%) or sometimes (20.75 \%) and the remaining minority rarely responding (18.87 \%) or never (1.89 \%) (Figure~\ref{f-98b8932ea974}). 

About patient's pain treatment expectations. They responded that when in too severe pain the majority responded that they sometimes (47.17 \%) usually (7.55 \%) or always (5.66 \%) think that the pain can never decrease; with a substantial minority never responding (28.3 \%) or rarely (11.32 \%) (Figure~\ref{f-b9d11dae0d4e}). However, a substantial majority responded that they were sometimes (33.96 \%), never (22.64 \%) or rarely (15.09 \%) afraid that something terrible could happen when they feel pain; with the minority usually responding (18.87 \%) or always (9.43 \%) (Figure~\ref{f-b6ce7dd3733e}). However, enquiring when their pain was an indication that they might be seriously ill, the responses were heterogeneous; with nearly equal proportions usually responding (22.64 \%), sometimes (22.64 \%), rarely (26.42 \%), or never (18.87 \%) with a very small group always responding (9.3 \%) (Figure~\ref{f-0ad4bf2bebd6}). On the same aspect, the majority never (49.06 \%), rarely (18.87 \%) or sometimes (22.64) thought that when their pain intensity increased, they might become paralyzed or more disabled. The minority always responded (7.55 \%) or usually (1.89 \%) (Figure~\ref{f-768e35ce19f6}). 

When questioned about whether their pain sensations were terrifying majority of the patients agreed, responding; sometimes (35.84 \%), usually (32.08 \%) or always (16.98\%); with minority disagreeing with never (5.66 \%) and rare (9.53 \%) responses (Figure~\ref{f-6676691511e1}). However, there seems to be uncertainty regarding the question as to whether the patients trembled when engaging in activities known to increases pain. In that case, a majority were unsure, sometimes responded (43.40 \%), with the remaining responses being split between never or rarely (13.21 \% or 20.75\%) and usually (16.98) or always (5.66 \%) (Figure~\ref{f-35da03c3ddec}).

On matters about the impact of their pain on their cardiological or respiratory outcome, the majority of the patients responded that sometimes (35.85 \%), rarely (30.19 \%) or never (5.66 \%) the pain cause their hearts to pound or race; with a substantial minority usually responding (18.87 \%) or always (9.43\%) or never (5.66 \%) (Figure~\ref{f-3cc45f2e4cd2}). These responses were quite similar to those regarding whether pain causes dizziness or fainting. In this case, a majority sometimes responded (41.51 \%) usually (18.87 \%) or always (5.66 \%), with a substantial minority responding never or rarely (16.98 \%) (Figure~\ref{f-72a173e6753a}). Similarly, the majority responded that the pain sometimes (30.19 \%), never or rarely (24.53 \%) makes them nauseous, with a minority usually responding (11.32 \%) or always (9.43 \%) (Figure~\ref{f-31023b0820c0}). Lastly, the majority of the patients responded that they sometimes (43.40 \%) find it difficult to calm down their bodies after periods of pain; with substantially higher proportion responding rarely (16.98 \%) or never (15.09 \%) compared to those responding usually (15.09 \%) or always (9.43 \%) (Figure~\ref{f-42d573a8cd90}).

In 2011 a study was conducted in Canada to measure the prevalence of chronic pain among the Canadian population, the study showed that the prevalence of chronic pain among adults (18+ years old) was 18.9\%. The study also indicated that the chronic pain prevalence was greater in older adults; females had a higher prevalence at older ages compared with males. Moreover, around half of those with chronic pain reported suffering for more than 10 years. Roughly one-third of those reporting chronic pain rated the intensity in the very severe range. The study also revealed that Lower Back Pain (LBP) was the most common site of chronic pain (CP), and arthritis was the most frequently named cause \unskip~\citep{582558:13491386}.

The study also concluded that the high prevalence of CP within adult populations in industrialized nations when the prevalence questions were carefully formulated and administered to large samples. Unfortunately, it also calls attention to the fact that a substantial proportion of Canadian adults live with CP that was longstanding and severe.

In July 2013, another study was revealed in California in USA, the study was measuring the Functional Disability among Chronic Pain Patients Receiving Long-Term Opioid Treatment, the study showed the roles of depression and important demographic characteristics in functional disability and employment status among chronic pain patients receiving long-term opioid therapy \unskip~\citep{582558:13491389}

The study showed that men had lower odds of having a functional disability compared to women. For every year increase in age, the odds that an individual has 30 or more days of limited activity increased by 2\% and the odds that he or she is unable to work due to health reasons increased by 4\%. Participants with at least some college education had 45\% lower odds of being unemployed due to health reasons compared to those with lower levels of education. Individuals with more depression symptoms (high depression) had higher odds of having a worse functional disability as gauged by all three measures \unskip~\citep{582558:13491389}. In April 2017, a study was carried out in Tehran in Iran to investigate the Chronic Pain in Chronic Heart Failure (HF). The study showed that chronic pain was a common symptom in patients with HF and had a remarkable impact on various aspects of the management modalities of these patients; optimal control of pain is impossible. Further investigations were needed to find a safe and efficacious way for controlling pain in patients with HF \unskip~\citep{582558:13491380}. Another study was conducted in Oxford University to estimate the Prevalence and Relevance of Pain in Older Persons, the study indicated that improved knowledge for both health professionals and patients, addressing the current research gaps and expansion of age-appropriate pain management services will be required to better meet the needs of our rapidly ageing population \unskip~\citep{582558:13491387}. 

The study also showed that pain was most common during the late-middle-aged phase of life (55{\textendash}65 years) and continues at approximately the same prevalence into older age (65+). This was true regardless of the anatomical site or the pathogenic cause of pain. The study mentioned that there was one exception appeared to be pain associated with degenerative joint disease (such as osteoarthritis) which showed an exponential increase until at least 90 years of age. Common age-associated conditions like dementia may result in a reduced frequency and intensity of pain. The study showed that daily pain was a major risk factor for developing a disability and the oldest age cohorts are most vulnerable. Discretionary and higher-order physical activities appear most affected, while basic activities of daily living may be modified but are rarely ceased altogether. Similar relationships have been documented for risk of depression and mood disturbance in older persons with persistent pain. Despite such well-characterized adverse impacts, pain often remains poorly treated in older persons. This occurs across all health care settings examined (emergency, acute, outpatient, long-term care) \unskip~\citep{582558:13491387}.

Another study was conducted in Texas in the USA, contributed to the literature in three main areas. The first contribution is the finding on the short amount of time spent on discussing chronic pain with older adults {\textendash} the median length is only 2.3 minutes (standard deviation= 3 minutes, minimum=6 seconds, maximum=15.4 minutes) for the combined talk time by physician and patient. When we separate out the time spoken by physician versus patient, we find that physicians spoke for less than a minute (0.8 minutes) whereas patients spoke for 1-minute \unskip~\citep{582558:13491388}. 

The second contribution was the high prevalence of chronic pain discussion among older patients in usual office visits in primary care {\textendash} 48\% of the visits involved at least one pain topic. While the third contribution was the priority given to discussing chronic pain {\textendash} 55\% of the chronic pain topics were raised during the first three topics of the visit with older patients. Despite the importance of chronic pain that could be inferred from the prevalence and the priority, the actual amount of time spent on chronic pain was quite minimal \unskip~\citep{582558:13491388}. 
    
\section{Conclusion}
Older adults in most parts of the world are vulnerable to chronic pains. The approaches employed to address pain among the aged should look beyond the capacity building to how good the existing and emerging programs can benefit those already in pain. Also, hospitals should partner with society to initiate community health programs for managing pain, especially among people aged above sixty years. Heath is an inevitable factor in determining the quality of life of an individual, irrespective of age or economic endowment. Therefore, progressive developments should be made in projects that focus on pain management. Essentially, there is no single approach known for treating pain; hence, stakeholders in the health sector can only embrace programs that focus on pain management. Moreover, the government, private sector, hospitals, and society should work together in developing policies and programs that can aid in the management of chronic pain. The elderly population is part of the community; therefore, the relevant stakeholders should ensure that this group of people enjoy a quality life that is free from chronic pain. 
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
